Author's response to reviews

Title: A presentation of minimally symptomatic large extradural hematoma in a patient with an arachnoid cyst: a case report and review of the literature

Authors:

Afsoun Seddighi (aseddighi@qums.ac.ir)
Amir Saied Seddighi (invincible19152@gmail.com)
Ali Reza Zali (dr_a_zali@yahoo.com)
Hesam Rahimi Baqdashti (baqdashti.hr@gmail.com)

Version: 6 Date: 4 November 2011

Author's response to reviews: see over
Dear Editor

The requested revisions are performed:

Please amend your title to the style of 'A presenting in B with C: a case report.'

Done

2. Within the first sentence of the presentation of the case: "The patient was a 23 year old Iranian worker who came to us and complained of ..." should be:
"The patient was a 23-year-old Iranian worker who came to us and complained of diffuse headache and several bursts of vomiting on the day of presentation."

Done

3. Please separate the "Patient consent" section from the "Competing Interests" section. (i.e., please make the two separate)

Ok, Done.

Sincerely Yours, Dr. Amir Saied Seddighi